Author's response to reviews

Title: Impact of physical fitness and biometric data on the quality of external chest compressions: a randomised, crossover trial

Authors:

Sebastian G Russo (s.russo@medizin.uni-goettingen.de)
Peter Neumann (neumann@ekweende.de)
Sylvia Reinhardt (Aivlys@gmx.net)
Arnd Timmermann (arnd.timmermann@helios-kliniken.de)
Andre Niklas (aniklas@gwdg.de)
Michael Quintel (mquintel@med.uni-goettingen.de)
Christoph B Eich (ceich@t-online.de)

Version: 6 Date: 6 September 2011

Author's response to reviews: see over
Title: Impact of physical fitness and biometric data on the quality of external chest compression - a randomised, cross-over trial

Göttingen, 06th September 2011

Dear Deesha Majithia,

Thank you very much for the opportunity to revise our manuscript. We also would like to thank the reviewer for his helpful comments.

We addressed the reviewer’s comments and provided $p$-values as well as the number of subjects within the tables, figures as well as the figure legends. All changes made in the manuscript are highlighted in yellow.

We hope to met your and the reviewer’s expectations and looking forward to your comments.

Best regards

Sebastian Russo